Title: Endobronchial Angiofibroma in the Aberrant Tracheal Bronchus Presenting as Spontaneous Pneumomediastinum

Authors:

Kyung Soo Kim (cskks@catholic.ac.kr)
Young Kyu Moon (mh-myk@hanmail.net)
Hyun Woo Jeon (ulnazoro@catholic.ac.kr)
Chan Beom Park (drcs5223@catholic.ac.kr)
Myeong Im Ahn (ami@catholic.ac.kr)
Kyo Young Lee (Leekoyo@catholic.ac.kr)
Jae Kil Park (Jaekpark@catholic.ac.kr)

Version: 5
Date: 30 April 2015

Author's response to reviews: see over
Reviewer's report:
Discretionary revisions:
could you please give more details on the technique used to repair tracheal laceration? Use of sealants?

Answer to the reviewer's comment:
Thank you for the comments. In order to repair the opened tracheal lesion after tumor removal, we simply sutured the lateral lesion of the trachea using multiple vicryl 4-0 sutures, as we have described in the manuscript. We didn't use any sealants after confirmation of no air leakage with saline test.

Following postoperative Chest Tomography (below) is showing a well-healed repaired tracheal lesion.

Thank you.